Author's response to reviews

Title: Pulmonary stenosis development and reduction of pulmonary arterial hypertension in atrioventricular septal defect

Authors:

emeline barth (emelinebarth@yahoo.fr)
hélène bouvaist (HBouvaist@chu-grenoble.fr)
gérard ninet (gerard.ninet@chu-lyon.fr)
gérald vanzetto (GVanzetto@chu-grenoble.fr)
stéphanie marlière (SMarliere@chu-grenoble.fr)

Version: 3 Date: 13 July 2009

Author's response to reviews:

revisions (answers to the reviewer LJ j Wagenaar) made in the main manuscript:

- Treatments until surgery: digitalo-diuretics which have been stopped in 1987.
- Surgeon closed the atrial and ventricular parts of the AVSD
- Vasoreactivity test = Notric oxide
- Figure 3: Proximal and distal pulomary artery diameter = 28 mm.